Author’s response to reviews

Title: Psychometric properties of the Chinese version of the Spiritual Care-Giving Scale (CSCGS) in nursing practice

Authors:

Yanli Hu (huyanli1212@126.com)

Lay Hwa Tiew (nurthl@nus.edu.sg)

Fan Li (lifan@jlu.edu.cn;870813123@qq.com)

Version: 1 Date: 08 Nov 2018

Author’s response to reviews:

Dear Alden Gross, Maria A. Fiatarone Singh, and Katerina Marcoulides,

Thank you very much for your precious comments and suggestions.

According to the comments and suggestions, we have revised the relevant part in the original manuscript. We also responded point-by-point to the comments of the editor and reviewers as listed below, all changes are marked in red in the revised version of the manuscript.

Thank you very much for your considering our manuscript for publication. We are looking forward to hearing from you soon.

With kindest regards,

Yanli Hu, Lay Hwa Tiew, Fan Li

Editor Comments:

Thank you for your revision. Please resubmit your manuscript, highlighting text in red font in the original submission. Because it is difficult to determine how you responded to Reviewer comments, please additionally restructure your response letter by listing a specific criticism from the reviewer (verbatim), then your response (to include all revisions to the manuscript text). Thank you sincerely.
Reviewer reports:

If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Response to Editor:

Editor Comments:
Thank you for your revision. Please resubmit your manuscript, highlighting text in red font in the original submission. Because it is difficult to determine how you responded to Reviewer comments, please additionally restructure your response letter by listing a specific criticism from the reviewer (verbatim), then your response (to include all revisions to the manuscript text). Thank you sincerely.

Response to Editor: Thank you very much for your encouragement and suggestions. We have resubmitted our manuscript, highlighting text in red font in the original submission and restructured our response letter by listing a specific criticism from the reviewer and our response. See the resubmitted manuscript (revisions marked in red) and the Response_to_Reviewers_and_Editors_comments and the R-response_to_Reviewers_and_Editors_comments.

Response to Reviewer reports:

Reviewer reports:
If improvements to the English language within your manuscript have been requested, you should have your manuscript reviewed by someone who is fluent in English. If you would like professional help in revising this manuscript, you can use any reputable English language editing service. We can recommend our affiliates Nature Research Editing Service (http://bit.ly/NRES_BS) and American Journal Experts (http://bit.ly/AJE_BS) for help with English usage. Please note that use of an editing service is neither a requirement nor a guarantee of publication. Free assistance is available from our English language tutorial (https://www.springer.com/gb/authors-editors/authorandreviewertutorials/writinginenglish) and our Writing resources (http://www.biomedcentral.com/getpublished/writing-resources). These cover common mistakes that occur when writing in English.

Response to Reviewer reports: Thank you very much for your precious suggestions. We have asked the English language editing service company AJE to improve the English language within our manuscript.

We tried our best to improve the manuscript and made some changes in the manuscript. And here we have list some changes and marked in red in revised paper. We appreciate for Editors/Reviews’ warm work earnestly, and hope that the correction will meet with approval.
Once again, thank you very much for your comments and suggestions.

Best regards,

Yanli Hu, Lay Hwa Tiew, Fan Li